\documentclass[letterpaper]{article}
%\documentclass[twocolumn,letterpaper]{article}
\usepackage{vmargin,natbib}
%\usepackage{times,mathptmx,vmargin,apalike}
\usepackage{graphicx}
\usepackage{hyperref}
\hypersetup{colorlinks=true,filecolor=blue,urlcolor=blue,breaklinks=true}
%
\usepackage{epsfig}
\setpapersize{USletter}
\setmarginsrb{1in}{1in}{1in}{1in}{0pt}{0mm}{0pt}{6mm}
\setlength{\columnsep}{0.25in}
\setlength{\parskip}{\baselineskip}
\setlength{\parindent}{0pt}
%\renewcommand{\familydefault}{phv}
%%\DeclareSymbolFont{symbols}{OMS}{psyr}{m}{n}
%%\SetSymbolFont{symbols}{normal}{OMS}{psyr}{m}{n}
\def\hang{\hangindent.25in \hangafter=1}
\newcommand{\Section}[1]{\vspace{-20pt}\section*{#1}\vspace{-10pt}}
\begin{document}

\centerline{\textsl{\textbf{Two-page Curriculum vitae}}}
\smallskip
\centerline{\Large\textbf{Forrest M. Hoffman}}
\centerline{Computational Earth Sciences Group}
\centerline{Oak Ridge National Laboratory (ORNL)}
\centerline{Oak Ridge, Tennessee 37831-6016}
\centerline{865-576-7680}
\centerline{\texttt{forrest@climatemodeling.org}}

\pagestyle{empty}

\Section{Education}

\hang
\textbf{M.S., Physics}, 2004.  University of Tennessee, Knoxville, TN \\
\textbf{Advisor:}  Dr. William E. Blass, Department of Physics and Astronomy \\
\textbf{Research Focus:}  Infrared spectroscopy and atmospheric spectroscopy \\
\textbf{Thesis:}  \textit{Analysis of Reflected Spectral Signatures and Detection of Geophysical Disturbance Using Hyperspectral Imagery}

\hang
\textbf{B.S., Physics}, 1991.  University of Tennessee, Knoxville, TN \\
\textbf{Advisor:}  Dr. William E. Blass, Department of Physics and Astronomy \\
\textbf{Research Focus:}  Astronomical photometry research and software for data reduction and analysis

\Section{Employment}

November 1993 -- present:  \textbf{Computational Climate Scientist}, joint appointments in the Computer Science \& Mathematics Division and the Environmental Sciences Division, Oak Ridge National Laboratory, Oak Ridge, Tennessee.

January 2002 -- October 2006: \textbf{Contributing Editor and Columnist}, ``Extreme Linux'' column, \textit{Linux Magazine}.

July 1992 -- November 1993:  \textbf{Scientific Programmer/Analyst}, Department of Geological Sciences, University of Tennessee, position at Environmental Sciences Division, Oak Ridge National Laboratory, Oak Ridge, Tennessee.

May 1989 -- June 1992:  \textbf{Knowledge Engineer}, Automated Sciences Group, Inc., position at Environmental Sciences Division, Oak Ridge National Laboratory, Oak Ridge, Tennessee.

September 1988 -- May 1989:  \textbf{Researcher Intern}, Oak Ridge Associated Universities (ORAU), position at Environmental Sciences Division, Oak Ridge National Laboratory, Oak Ridge, Tennessee.

1987 -- September 1988:  \textbf{Observer/Telescope Operator}, High Altitude Observatory, National Center for Atmospheric Research, position at Mauna Loa Solar Observatory, Hilo, Hawai'i.

\Section{Honors and Awards}

ORNL Outstanding Mentor Award.  Oak Ridge National Laboratory and Oak Ridge Associated Universities, February 2008.

Significant Event Award for contribution to NSF's National Ecological Observatory Network (NEON) Design Committee.  Oak Ridge National Laboratory/UT-Battelle, LLC, March 2006.

Outstanding Paper in the Discipline of Landscape Ecology. W. W. Hargrove,
F. M. Hoffman and P. M. Schwartz, ``A fractal landscape realizer
for generating synthetic maps,'' \textit{Conserv. Ecol.} 6(1):2
(2002). Awarded by the International Association for Landscape Ecology,
U. S. Regional Chapter, April 2004.
%
% World-Class Teamwork Award. Oak Ridge National Laboratory Values
% Committee, October 19, 1999.
%
%President's Award for Continuous Improvement. Oak Ridge National
%Laboratory/Lockheed Martin Energy Research Corp., November 1997.
%
%ORNL Technical Achievement Award. Oak Ridge National Laboratory/Lockheed
%Martin Energy Research Corp., 1996.

%Distinguished Achievement Award for Operational Support. Oak Ridge
%National Laboratory, Environmental Sciences Division, August 1995.
%
%Significant Event Award. Martin Marietta Energy Systems, June 1995.

\nocite{Kloster_Biogeosci_20100611,
Randerson_GCB_20091001,
Erickson_JGR_20080315,
Hoffman_JPConf_20071201,
Dickinson_JClim_20060601,
Hargrove_JGS_20060701,
Hoffman_IJHPCA_20050801,
Hoffman_EI_20050803,
Hargrove_LandscapeEcol_20050501,
White_GRL_20050218,
Saxon_EcolLett_20050101,
Hargrove_EnvironManage_20040401,
Hargrove_ConservEcol_20020213,
Hargrove_SciAm_20010801}

\renewcommand\refname{Selected Publications}
\bibliographystyle{unsrtnat}
\bibliography{abbrev,hoffman,linux-mag}

\end{document}
